Author’s response to reviews

Title: Self-motivated medical care-seeking behaviors and disease progression in a community-based cohort of chronic hepatitis B virus-infected patients in China

Authors:

Hui Zheng (zhenghui@chinacdc.cn)
Guomin Zhang (zhanggm@chinacdc.cn)
Fuzhen Wang (wangfz@chinacdc.cn)
Zundong Yin (yinzd@chinacdc.cn)
Miao Ning (miaoning@chinacdc.cn)
Xiaojin Sun (sunxj@chinacdc.cn)
Xiaofeng Liang (laingxf@chinacdc.cn)
Fuqiang Cui (nip2018@163.com)

Version: 1 Date: 22 Oct 2018

Author’s response to reviews:

1. Please can you clarify whether you obtained written informed consent from participants to participate in this study and include these details in your manuscript.

Yes, we began to conduct the questionnaire investigation after the participants signed the informed consent form. and this has already been added in the manuscript.

2. Upon resubmission, please make sure that your manuscript contains the following seven sections under the heading 'Declarations' at the end of the main text.

Done